Author's response to reviews

Title: Lyme neuroborreliosis in HIV-1 positive patients successfully treated with oral doxycycline: case series and literature review

Authors:

Daniel Bremell (daniel.bremell@infect.gu.se)
Christer Säll (christer.sall@vgregion.se)
Magnus Gisslén (magnus.gisslen@infect.gu.se)
Lars Hagberg (lars.hagberg@medfak.gu.se)

Version: 3 Date: 22 July 2011

Author's response to reviews: see over
Cover Letter

Second revision “Lyme neuroborreliosis in HIV-1 positive patients successfully treated with oral doxycycline: case series and literature review”

Here is a point-by-point description of changes made in response to the concerns raised by the editorial team:

- In the Abstract, gender, ethnicity, and age of the patients has been added.
- In Case Presentations, all dates have been replaced by appropriate time references.
- The Method and Material section has been removed.
- All hospital names have been removed.

Gothenburg 2011-07-22

Daniel Bremell
Christer Säll
Magnus Gisslén
Lars Hagberg